Author's response to reviews

Title: Association between gestational age at birth, antenatal corticosteroids, and outcomes at 5 years: Multiple Courses of Antenatal Corticosteroids for Preterm Birth Study at 5 years of age (MACS-5)

Authors:

Elizabeth V Asztalos (elizabeth.asztalos@sunnybrook.ca)
Andrew R Willan (andy@andywillan.com)
Kellie E Murphy (kmurphy@mtsinai.on.ca)
Stephen G Matthews (stephen.matthews@utoronto.ca)
Arne Ohlsson (aholsson@mtsinai.on.ca)
Saroj Saigal (saigal@mcmaster.ca)
B Anthony Armson (Anthony.Armson@iwk.nshealth.ca)
Edmond N Kelly (ekelly@mtsinai.on.ca)
Marie-France Delisle (mdelisle@cw.bc.ca)
Amiram Gafni (gafni@mcmaster.ca)
Shoo K Lee (sklee@mtsinai.on.ca)
Renee Sananes (renee.sananes@sickkids.ca)
Joanne Rovet (jrovet@sympatico.ca)
Patricia Guselle (patricia.guselle@sickkids.ca)
Kofi Amankwah (Kofi.Amankwah@mackenziehealth.ca)

Version: 7 Date: 28 July 2014

Author's response to reviews: see over
July 28, 2014

To whom it may concern:

Re: Association between gestational age at birth, antenatal corticosteroids, and outcomes at 5 years: Multiple Courses of Antenatal Corticosteroids for Preterm Birth Study at 5 years of age (MACS-5)

Thank you for providing us the opportunity to provide additional revisions for the above-entitled manuscript.

Below are our “point-by-point” responses to the revisions requested in the editorial comments:

1. Please provide a figure legend for Figure 1 in the main MS file according to IforA.
   The figure legend now has the figure number and a short title of the figure in sentence case..

We hope that we have addressed all the revisions appropriately and adequately and that the revised manuscript meets with editorial approval.

With kind regards,

Elizabeth Asztalos

Elizabeth Asztalos MD MSc FRCPC